

# The Role of Irrigation Expansion on Historical Climate Change: Insights From CMIP6

A Al-yaari, Agnès Ducharne, W Thiery, F Cheruy, D Lawrence

# ► To cite this version:

A Al-yaari, Agnès Ducharne, W Thiery, F Cheruy, D Lawrence. The Role of Irrigation Expansion on Historical Climate Change: Insights From CMIP6. Earth's Future, 2022, 10 (11), 10.1029/2022ef002859. hal-03866215

# HAL Id: hal-03866215 https://hal.science/hal-03866215

Submitted on 22 Nov 2022

**HAL** is a multi-disciplinary open access archive for the deposit and dissemination of scientific research documents, whether they are published or not. The documents may come from teaching and research institutions in France or abroad, or from public or private research centers. L'archive ouverte pluridisciplinaire **HAL**, est destinée au dépôt et à la diffusion de documents scientifiques de niveau recherche, publiés ou non, émanant des établissements d'enseignement et de recherche français ou étrangers, des laboratoires publics ou privés.

23284277, 2022, 11, Dov

/agupubs.onlinelibrary.wikey.com/doi/10.1029/2022EF002859 by Sorbonne Università, Wiley Online Library on [22/11/2022]. See the Terms and Conditions (https://onlinelibrary.wikey.com/terms-and-conditions) on Wikey Online Library for rules of use; OA



# **Earth's Future**

# **RESEARCH ARTICLE**

10.1029/2022EF002859

#### **Special Section:**

CMIP6: Trends, Interactions, Evaluation, and Impacts

#### **Key Points:**

- Trends of six essential climate variables were compared between Coupled Model Intercomparison Project phase 6 (CMIP6) simulations with (CMIP6.irr) and without (CMIP6.noirr) irrigation
- CMIP6.irr exhibits a distinct behavior in better capturing the upward trend of observed latent heat flux over heavily irrigated regions
- Over the same regions, while CMIP6. irr exhibits a cooling trend, CMIP6. noirr exhibits a warming trend

#### **Supporting Information:**

Supporting Information may be found in the online version of this article.

#### Correspondence to:

A. Al-Yaari, amen.alyaari@gmail.com

#### Citation:

Al-Yaari, A., Ducharne, A., Thiery, W., Cheruy, F., & Lawrence, D. (2022). The role of irrigation expansion on historical climate change: Insights from CMIP6. *Earth's Future*, *10*, e2022EF002859. https://doi.org/10.1029/2022EF002859

Received 17 APR 2022 Accepted 29 SEP 2022

© 2022. The Authors. Earth's Future published by Wiley Periodicals LLC on behalf of American Geophysical Union. This is an open access article under the terms of the Creative Commons Attribution License, which permits use, distribution and reproduction in any medium, provided the original work is properly cited.

# The Role of Irrigation Expansion on Historical Climate Change: Insights From CMIP6

A. Al-Yaari<sup>1</sup>, A. Ducharne<sup>2,3</sup>, W. Thiery<sup>4</sup>, F. Cheruy<sup>3,5</sup>, and D. Lawrence<sup>6</sup>

<sup>1</sup>Univ. Grenoble Alpes, IRD, CNRS, Grenoble INP, Insitut des Geosciences de l'Environnement (IGE, UMR 5001), Grenoble, France, <sup>2</sup>Sorbonne Université, UMR 7619 METIS, Case 105, Paris, France, <sup>3</sup>IPSL (Institut Pierre Simon Laplace), Sorbonne Université, CNRS, Paris, France, <sup>4</sup>Department of Hydrology and Hydraulic Engineering, Vrije Universiteit Brussel, Brussels, Belgium, <sup>5</sup>LMD (Laboratoire de Météorologie Dynamique), Sorbonne Université, ENS, IPSL Université, École polytechnique, Institut Polytechnique de Paris, CNRS, Paris, France, <sup>6</sup>National Center for Atmospheric Research, Boulder, CO, USA

**Abstract** To produce food for a growing world population, irrigated areas have increased from approximately 0.63 million km<sup>2</sup> of land in 1900 to 3.1 million km<sup>2</sup> of land in 2005. Despite this massive expansion, irrigation is still overlooked in most state-of-the-art Earth system models (ESMs) involved in the Coupled Model Intercomparison Project phase 6 (CMIP6). To our knowledge, only three CMIP6 models represent irrigation activities: CESM2, GISS-E2-1-G, and NorESM2-LM. Here, we investigate the role of irrigation on historical climate at global and regional scales by analyzing trends of key surface climate variables in CMIP6 simulations during 1900-2014. The three models including irrigation show distinct behavior from the 15 models without irrigation over intensively irrigated areas (i.e., >50% of grid cell area is equipped by irrigation): both annual (months that correspond to monthly air temperature higher than 274 K) mean latent heat flux (LHF) and soil moisture increase over time, in contrast to the models without irrigation that show no trend or even a negative trend. The positive LHF trend over intensively irrigated areas in the irrigation-on models is consistent with three satellite-based LHF products. While annual (considering the warmest month in a year) warming trends are found in these regions for most of the no-irrigation models, the increase in LHF induces a cooling trend for the models with irrigation, which was not confirmed by observational air temperature data sets. These findings, supported by satellite data, indicate the importance of improved representation of land management in the next generation of ESM.

**Plain Language Summary** Expansion of irrigated areas influences long-term terrestrial water and energy cycles, primarily via enhanced evapotranspiration from irrigated surfaces. Here, we analyze long-term (1900–2014) trends of latent heat flux (LHF), soil moisture, precipitation, net radiation, and air temperature from the latest generation of Earth system models (ESMs) involved in the Coupled Model Intercomparison Project phase 6 (CMIP6). Only three CMIP6 models account for irrigation activities, while all others omit irrigation. Based on the trend results, a distinct behavior of the three models is observed, with the annual mean LHF and soil moisture generally increasing over heavily irrigated regions, with a corresponding cooling trend in the annual maximum of monthly air temperature. We find that neglecting irrigation reduces the accuracy of LHF trend estimates over heavily irrigated regions, as the performance of models without irrigation as a historical climate forcing and underlines the benefit of incorporating irrigation in the next generation of ESM to reduce uncertainties in historical climate simulations and enhance the reliability of future projections.

# 1. Introduction

Approximately 70% of the world's total land surface is under human management (Luyssaert et al., 2014), which is altering Earth's landscape. Human–water interactions (Oki, 1999), in particular, combined with increasing population growth (Vörösmarty et al., 2000), are expected to further exacerbate hydrologic extremes such as floods and droughts (Milly et al., 2002). Societies are increasingly intervening in the hydrological cycle, for instance, through irrigating cultivated areas, which is likely to expand in the future to meet the growing demand for food and energy (Foley et al., 2011). According to Puy (2018), irrigated areas have increased proportionally faster than the growth of the world population. Crops grown in irrigated areas account for over 40% of global food production (Abatzoglou et al., 2013). Irrigated areas represent ~90% of consumptive water use globally (Rost

governed by the applicable Crea

et al., 2008), and have increased fivefold from approximatively 0.63 million km<sup>2</sup> of land in 1900 to 3.1 million km<sup>2</sup> of land in 2005 (Siebert et al., 2013). Irrigation induces biogeochemical and biogeophysical effects on climate (Mahmood et al., 2014; Sacks et al., 2009), and its expansion can alter precipitation patterns at the regional (DeAngelis et al., 2010) and global scales (Guimberteau et al., 2012; Puma & Cook, 2010). Irrigation amounts per unit irrigation area have been increasing over time just like irrigation extent has been Increasing (Siebert et al., 2015). Thiery et al. (2017) have shown that irrigation quantities are applied preferentially during hot (Mediterranean) or dry (South Asia) days.

The additional transpiration and evaporation associated with irrigation tend to cool the local climate (Yang et al., 2020). Moreover, irrigation perturbs the local surface energy and water budgets (Haddeland et al., 2006; Ozdogan et al., 2010; Pokhrel et al., 2011) and modulates local to regional atmospheric processes and landatmosphere interactions (Qian et al., 2020) via the impact of irrigation on the partitioning between sensible and latent heat fluxes (LHFs; Bonfils & Lobell, 2007; Kanamaru & Kanamitsu, 2008; Saeed et al., 2009; Sorooshian et al., 2014; Yang et al., 2020). Irrigation affects water resource availability over heavily populated irrigation hotspots (Vicente-Serrano et al., 2019). Despite the clear climate and water cycling impacts of historical and future irrigation (de Vrese & Stacke, 2020; Hurtt et al., 2020; Lawrence et al., 2016), it has generally not been considered in evaluations of drivers of observed climate change (Thiery et al., 2020). The role of irrigation on land-atmospheric coupling and land surface model initialization has been demonstrated in several studies (H. Kim & Lakshmi, 2019; Lawston et al., 2015; Lu et al., 2015, 2017; Pryor et al., 2016). Lu et al. (2017) found that irrigation led to a significant decline in coupling strength between the irrigated land and the atmosphere in the Midwestern United States. Lu et al. (2015) showed that the inclusion of realistic irrigation physics led to a better description of the land-atmospheric coupling. Lawston et al. (2015) demonstrated that the inclusion of irrigation in land surface models had the potential to improve forecasts, providing a better tool to adapt to increasing crop demands.

Earth observations and Earth system models (ESMs) have been widely used to study the past and potential future evolution of the Earth system processes that determine the exchanges between land surfaces and the atmosphere (Sellers et al., 1997). Incorporating irrigation among other land management practices (e.g., conservation agriculture, reservoir operation, and groundwater pumping) into ESMs has been highlighted as an important research direction (Hirsch et al., 2018; Mizuochi et al., 2021; Pokhrel et al., 2016). Despite intensive improvements in the representation of soil and vegetation processes in land models in ESMs (Cheruy et al., 2020), and although changes in the water cycle due to land and water management are now of global importance, most state-of-the-art ESMs do not yet account for irrigation as they do for land cover change, aerosols, and other forcings (Cook et al., 2015).

Consequently, persistent biases in a number of key ESM variables (e.g., soil moisture and temperature) were observed in several regions (Al-Yaari et al., 2019; Cheruy et al., 2014). The existence of systematic hot and dry biases (precipitation deficit) in summer is, for instance, a known deficiency of many models participating in Phase 5 of the Coupled Model Intercomparison Project (CMIP5; Cheruy et al., 2014), particularly in regions of transition between humid and dry climates, where surface/atmosphere interactions contribute to the dryness of the future climate (Berg et al., 2014). This deficit could be due to the difficult characterization of the variables of the critical zone (Lahoz & de Lannoy, 2014), surface-atmosphere couplings (Cheruy et al., 2014), or, as mentioned above, the poor representation (or even an absence of representation) of irrigation activities in space and time in land modules within the ESMs (Qian et al., 2020).

Compared with that in CMIP5, the number of models participating in CMIP phase 6 (CMIP6) has increased greatly, and most models used an updated version (Cook, Mankin, et al., 2020; Danabasoglu et al., 2020; Hajima et al., 2020; Müller et al., 2018; Swart et al., 2019; Wu et al., 2019). However, as in CMIP5, most ESMs included in CMIP6 still neglect irrigation activities in climate projections. This negligence of irrigation activities in land models is due to a number of challenges (Thiery et al., 2017, 2020); for instance, the applied amount of irrigation water needs to be realistic, and there is uncertainty in the timing, techniques, and efficiency of irrigation.

However, progress has been increasingly made in this line of research to characterize changes in climate in response to irrigation. For instance, the Land Use Model Intercomparison Project (Boysen et al., 2020; Lawrence et al., 2019) was launched recently within the framework of CMIP6 to assess the biogeophysical effects of land use and land cover change (LULCC) in a multimodel framework. The majority of previous studies quantified

the impact of irrigation on weather, climate, and hydrology at local and regional scales based on a single model (Cook et al., 2015; Gormley-Gallagher et al., 2020; Thiery et al., 2017), focusing on temperature extremes (Hauser et al., 2019; Hirsch et al., 2017), or focused on incorporating irrigation into land-only hydrological models (Pokhrel et al., 2016).

To the best of our knowledge, no previous study has investigated the role of irrigation on historical climate using an ensemble of ESMs. Here, we take advantage of the newly available CMIP6 simulations from the most recent generation of climate models to re-examine the impact of irrigation activities at the global scale by investigating long-term trends of five essential climate variables (ECVs) that characterize the Earth's near-surface climate (evapotranspiration [ET], or the corresponding surface energy term, the LHF; precipitation; soil moisture; radiation components; and air temperature) over the 1900–2014 period. Within this study, we (a) compare models including and excluding irrigation (hereinafter referred to as CMIP6.irr and CMIP6.noirr, respectively) using CMIP6 historical (fully coupled) simulations; and (b) evaluate simulated trends against existing observational data sets.

## 2. Materials and Methods

#### 2.1. Trend Analyses

For all CMIP6 variables except air temperature, we computed for each year the mean of the months with a monthly mean temperature higher than 274 K, assuming it approximately captures the growing season (Jiang et al., 2017; Prentice et al., 2011), when irrigation can be applied. One bias that the models could have is that irrigation quantities, even though realistic at the annual timescale (Thiery et al., 2017), might be applied in different months/seasons than the ones where irrigation is actually happening (Jha et al., 2022). For the air temperature, the annual maximum of monthly air temperature ( $T_{max}$ ) was considered instead, as previous studies have shown more significant irrigation impacts on hot extremes (Chen & Dirmeyer, 2020; Pitman et al., 2012; Thiery et al., 2020).

The impact of irrigation on the average of fluxes has already been studied for the models with irrigation versus their versions without irrigation (Cook et al., 2015; Thiery et al., 2017), so here we focus on the trends. To investigate the impact of irrigation on historical climate, the magnitude, or the rate at which trends are increasing/ decreasing is compared between CMIP6.irr and CMIP6.noirr within the 1900–2014 period. The widely used hydrometeorological time series (Tabari et al., 2011) Sen's slope nonparametric estimator (Sen, 1968) is used to estimate the magnitude of the trend (change per unit time) of the areas equipped for irrigation as well as the CMIP6 ECVs. The concept is to calculate the slope between all the pairs of ordered time points and then take the median of all these slopes as a unique estimator, representative of the entire time series. For a sample of N pairs:

$$S_k = \frac{x_i - x_j}{i - j} \text{ for all } i > j \text{ and } k = 1, \dots, N,$$

$$\tag{1}$$

where  $x_i$  and  $x_j$  denote the corresponding data values at times *i* and *j*. The *N* values of  $S_k$  are ranked from smallest to largest, and the median of Sen's slope is calculated as follows:

$$S_{\text{med}} = \begin{cases} S_{(N+1)/2} & \text{if } N \text{ is odd} \\ \frac{1}{2} \left( s_{N/2} + S_{(N+2)/2} \right) & \text{if } N \text{ is even} \end{cases}$$
(2)

A positive  $S_{med}$  represents an increasing trend, and a negative sign represents a decreasing trend. Its value gives a quantitative estimate of the trend of the time series.

This test has low sensitivity to abrupt breaks due to inhomogeneous data series, does not require the assumption of normality, is robust against outliers, and can handle missing values (Chandler & Scott, 2011). The nonparametric Mann-Kendall test (Kendall, 1957) was applied to assess the strength and significance level of the trend. The trend is considered statistically significant (i.e., a significantly decreasing or increasing trend) if the *p*-value <0.05.





**Figure 1.** (a) Map of global FAO irrigated areas circa 2005 on a  $1^{\circ}$  by  $1^{\circ}$  grid, shown as a percentage of the land area in each  $1^{\circ}$  pixel (Siebert et al., 2013). (b) The evolution in time of the Historical Irrigation Dataset (HID) areas equipped for irrigation (left axis) and the total area (gray line and on right axis) according to the classes of the percentage of the land area in each  $1^{\circ}$  pixel circa 2005 (Siebert et al., 2015). (c) Spatial average (solid blue line) of pixels with significant (*p*-value <0.05) Sen's slope values per decade of the HID equipped areas for irrigation calculated between 1950 and 2005 with a standard deviation as a measure of uncertainty (light blue envelope). The second black line in Figure 1c and on right axis represents the percentage of grid cell areas of global land.

#### 2.2. Irrigation Data Sets

To study historical irrigation expansion, we use the Historical Irrigation Dataset (HID) version 1.0 (Siebert et al., 2015). This data set provides gridded maps of the area equipped for irrigation expressed in hectares per grid cell based on national and subnational statistics at a 5 arc-minute resolution and a temporal resolution of 10 years from 1900 to 1980, increasing to 5 years from 1985 to 2005.

In addition, we use the latest version (5.0) of the FAO "Global Map of Irrigation Areas" (Siebert et al., 2013). This map (Figure 1a) quantifies the areas equipped for irrigation circa 2005 at a 5 arc-minute resolution (equal to 0.083°, i.e., ~8 to 9 km approximately). It is a static map of irrigated areas, which are expressed as the fraction of each 5 arc-minute pixel area that is equipped for irrigation. This map was developed through a collaboration between the United Nation's FAO Land and Water Division and the Rheinische Friedrich-Wilhelms-Universität in Bonn, Germany. It relies on the data collected at national and sub-national level about irrigation water with-drawal per year (m<sup>3</sup>), and is distributed via FAO's global water information system, AQUASTAT. The FAO "Global Map of Irrigation Areas" has been used as input to most continental to global scale models accounting for irrigation (e.g., de Rosnay et al. (2003), with an earlier version), and serves as a basis and/or benchmark for many recent studies aiming at proposing new irrigation maps (Meier et al., 2018).

It should be noted that this map is not exactly the same as the map in the historical irrigation data set for 2005 (HID). To compile the FAO map, the best and most precise estimate available for a year circa 2005 was used, whereby the reference year differs across countries. In the HID data set, in contrast, the authors tried to exactly match the situation in 2005 (see Figure S1 in Supporting Information S1 for the HID and FAO maps circa 2005). The spatial values were classified into six levels at a 1° by 1° resolution, as illustrated in Figure 1a. The approximate fraction of the number of land pixels for each class (excluding Greenland and Antarctica) is illustrated in Figure 1c (right axis).

The evolution of the areas equipped for irrigation is illustrated in Figure 1b throughout the 1900–2005 period for each irrigation class (Figure 1a). Since 1950, there has been an abrupt increase in irrigation areas, mainly for

grid-cell areas equipped for >5% irrigation. Figure 1c (left axis) shows the significant Sen's slope in the HID irrigation data sets: all irrigated areas expanded from 1900 to 2005, with the highest trends occurring in areas that were heavily irrigated. In irrigated grid cells where the irrigated grid cell fraction is greater than 50%, there is a rate of increase of 542 ha (spatially averaged) per decade from 1950 to 2005. In the subsequent analysis, the influence of this increasing trend on climate variables is presented and discussed. We interpolate the irrigation data sets onto 1° by 1° grid using bilinear interpolation (e.g., Stanfield et al., 2016).

#### 2.3. CMIP6 Climate Simulations

#### 2.3.1. Overview of CMIP6 Historical Simulations

CMIP6 is a primary source of data for climate change studies. In this study, historical simulations from 18 fully coupled climate models (listed in Table S1 in Supporting Information S1; constrained by data availability) were downloaded from the Earth System Grid Federation (ESGF) database (available at https://esgf-node. llnl.gov/search/cmip6/). These simulations were run under standardized protocols with uniform output and a consistent structure (Eyring et al., 2016). In this study, we use historical fully coupled (ocean-atmosphere-land) runs that are driven as much as possible by historical forcing data sets based on observations over 1850–2014 (Eyring et al., 2016). They include time-evolving data sets of atmospheric composition, solar forcing, and gridded LULCC, following the Land-Use Harmonization database (LUH2; Hurtt et al., 2020). However, not all ESMs represent the effects of LULCC with the same level of complexity, and most models are not able to make full use of the information provided in the LULCC data sets produced for CMIP6 (Hurtt et al., 2020; Lawrence et al., 2016).

The CMIP6 historical simulations start in 1850 until 2014, from which we examined the 1900–2014 period, which overlaps with the availability of historical irrigation data sets. These simulations with and without irrigation will be hereinafter referred to as CMIP6.irr and CMIP6.noirr, respectively. Historical simulations by CMIP6 models offer several members, and we focus on a single member for each model, the ones archived as "r1i1p1f1" within the ESGF databases, that is, with realization number 1, initialization method indicator 1, perturbed physics number 1, and forcing number 1. Other ensemble members were used for some models to check the internal variability among the different members.

To address the impact of irrigation on the simulated climate, we investigate five ECVs: soil moisture (hereinafter referred to as SM; in mm), annual maximum of monthly air temperature ( $T_{max}$ ; in K), LHF (in W/m<sup>2</sup>), precipitation (hereinafter referred to as Pr; in mm/day), and net radiation (hereinafter referred to as  $R_{net}$ ; in W/m<sup>2</sup>). Net radiation is computed as the balance of four radiation components (all in W/m<sup>2</sup>), as follows:  $R_{net} = SW_{net} + LW_{net} = (SW_{down} - SW_{up}) + (LW_{down} - LW_{up})$ . LW<sub>net</sub> is the net longwave surface radiation, and SW<sub>net</sub> is the net shortwave surface radiation, with down and up referring to surface downwelling and upwelling radiation, respectively. We focus on monthly means for SM, LHF, Pr, and  $R_{net}$  and on the annual maximum of air temperature ( $T_{max}$ ) and interpolate all data over the 1900–2014 period onto a common 1° by 1° grid using bilinear interpolation (e.g., Stanfield et al., 2016).

#### 2.3.2. Main Features of the CMIP6.irr Models

Among the 18 selected climate models (Table S1 in Supporting Information S1), only three of them, namely CESM2, GISS-E2-1-G, and NorESM2-LM, include time-varying irrigation. These three CMIP6.irr models all respond to prescribed time-varying irrigated areas, which are prescribed as input data from the CMIP6 LULCC database (Hurtt et al., 2020). They were taken from the HYDE 3.2 database (Goldewijk et al., 2017), where irrigation is based on the HID data set (Siebert et al., 2015), shown in Figure S1 in Supporting Information S1. Therefore, the delineation of the areas where irrigation is occurring (current extent as well as historical expansion) is documented in the three CMIP6 models from the best available data sets.

CESM2 and NorESM2-LM use the same Community Land Model version 5 (CLM5) (Lawrence et al., 2019) except for one specific modification made to the surface water treatment in NorESM2-LM: the wetland unit in earlier versions of CLM was replaced by a surface water pool. This new feature does not have a frozen state, but is added to the snow pack when frozen, as stated by Seland et al. (2020). The realistic simulation of land surface fluxes in agricultural regions imposes to describe the dynamics of crop growth and its control by land management practices (Lawrence et al., 2016; Lu et al., 2015), as it is possible with CLM5. The crop types are corn

(tropical and temperate varieties), soybean (tropical and temperate), wheat, rice, sugarcane, and cotton. The leaf area index (LAI) of crops and natural vegetation are simulated prognostically (dynamic), and crop management includes sowing, harvest, industrial fertilization and irrigation. The performance of the CLM5 crop model was evaluated by Lombardozzi et al. (2020).

While CESM2 uses the Community Atmospheric Model version 6 (Danabasoglu et al., 2020), NorESM2 uses CAM6-Nor with some specific modifications made to the atmospheric component: an independent module for the life cycle of particulate aerosols, changes in the moist convection scheme and the local moist energy formulation, and a representation of aerosol-radiation-cloud interactions, as stated by Seland et al. (2020). Readers are directed to Seland et al. (2020) for more information about more differences between CESM2 and NorESM2-LM. GISS-E2.1-G is the coupled atmosphere-ocean-sea-ice-land NASA Goddard Institute for Space Studies (GISS) ESM (model E2.1), with *G* referring to the "GISS Ocean" model, which is an updated and improved version of GISS-E2-R used in CMIP5 (Kelley et al., 2020; Miller et al., 2021). GISS-E2.1-G describes three types of crops (Y. Kim et al., 2015), but it does not have dynamic vegetation, and all the parameters describing vegetation structure, including LAI, are prescribed at a monthly time step LAI (Cook, McDermid, et al., 2020).

As described by Cook, McDermid, et al. (2020) and Miller et al. (2021), irrigation-as the water required to ensure optimal crop growth—in GISS-E2-1-G is prescribed from an updated version of the time-varying irrigation water demand (IWD) data set of Wisser et al. (2010). These IWD data were generated by combining the best empirically constrained global state-of-the-art estimates of irrigation applications: a land-only terrestrial water balance model, the University of Frankfurt/FAO Global Maps of Irrigated Areas, and crop-specific calendars and water demand coefficients, which consider regional cropping practices. The IWD is calculated between 1900 and 2005 using the irrigation areal extent from Siebert et al. (2015) and is estimated between 1871 and 1900 by extrapolating backwards from the trend between 1900 and 1929. Irrigation is held constant after 2005 using the flux from 2004. Irrigation water to satisfy the irrigation demand is first sourced from lakes and rivers within the same irrigated grid cell. However, if the prescribed IWD cannot be fully satisfied by this flux, additional water is added from fossil groundwater, which represents an addition to the total mass of water within GISS-E2-1-G. Shukla et al. (2014) demonstrated that the simulated variability of summer monsoon circulation indices was improved by the inclusion of irrigation activities in GISS-E2-1-G. Nevertheless, irrigation in GISS-E2-1-G is applied on the entire vegetated grid cell fraction instead of only the crop fraction, causing all vegetation to respond to irrigation, which may lead GISS-E2-1-G to be too sensitive to irrigation, thus amplifying ET and the subsequent climate response.

Unlike GISS-E2-1-G, irrigation is internally calculated in CESM2 and NorESM2-LM. The irrigated crop water demand is triggered by a soil moisture deficit when there are growing crops (Lombardozzi et al., 2020; Sacks et al., 2009). The water for irrigation is initially sourced from river water storage (Danabasoglu et al., 2020); however, if this source is not fully sufficient for the demand, water is pulled diffusely from the oceans (Lawrence et al., 2018, 2019). Similarly to GISS-E2-1-G, this methodology effectively assumes that irrigation is never water-limited, which would be the case in some situations in reality. Note also that the timing of irrigation is tied to the crop life cycle, which is prognostic and known to be biased in CLM5 in some regions, especially over India, where crop planting can occur too early in the year during the dry season (Lombardozzi et al., 2020). Furthermore, the irrigation parameterization does not distinguish between irrigation techniques (e.g., drip vs. flood), which may further bias the results. Thiery et al. (2017) evaluated the simulated irrigation amounts within CESM1.2.2. Locally, they may be overestimated or underestimated compared to observations and hence may overestimate or underestimate the climate response to irrigation. At the regional scale, the simulated irrigation amounts show a reasonable match with observed rates, with higher skills in South Asia. Most current irrigation implementations take a certain amount of water and apply it to the surface (either above or below the canopy). This water can either infiltrate, evaporate or run off. But in case of rice paddies, for instance, small basins are created where the water is actually stagnating. Current ESMs do not capture this effect at the moment, and therefore likely underestimate the effect of paddy irrigation on near-surface climate.

A common feature of the three climate models is that they do not consider groundwater as a possible water source for irrigation via groundwater pumping. In CESM2 and NorESM2-LM, irrigation is never limited by water availability, and if water needed for irrigation exceeds what is available in rivers, it is drawn from the oceans to ensure water conservation in the coupled climate system. Such unlimited irrigation is a way to circumvent the lack of groundwater pumping, since it helps achieving high irrigation rates even where groundwater is heavily used.

Felfelani et al. (2021) recently developed a prognostic groundwater module for CLM5, accounting for aquifer pumping, lateral groundwater flow, and conjunctive use of groundwater for irrigation, and it will probably be incorporated in a future CESM version. Moreover, while dynamic lake area and dam management have recently been implemented in CLM5, these parameterizations currently do not influence irrigation water availability in the model (Vanderkelen et al., 2021).

#### 2.4. Observation-Based ECVs

The observation-based products listed below serve as a benchmark between the two different CMIP6 categories: CMIP6.irr and CMIP6.noirr. These products were interpolated to the 1° by 1° grid used for the CMIP6 models.

#### 2.4.1. GLASS

The Global Land Surface Satellite (GLASS) ET product is estimated using the Bayesian model averaging method (Yao et al., 2014). This approach merges five process-based ET algorithms, including the Moderate Resolution Imaging Spectroradiometer (MODIS) ET product algorithm, the remote-sensing-based Penman-Monteith ET algorithm, the Priestley-Taylor-based ET algorithm, the modified satellite-based Priestley-Taylor ET algorithm, and the semiempirical Penman ET algorithm of the University of Maryland. The GLASS ET products cover the 1982–2015 period with a spatial resolution of  $0.05^{\circ}$  by  $0.05^{\circ}$  and an 8-day period. GLASS ET is provided in mm/ month, which is almost equivalent to W/m<sup>2</sup> (as one needs ~30 W/m<sup>2</sup> to evaporate 1 kg/m<sup>2</sup> of water).

#### 2.4.2. GLEAM

The Global Land Evaporation Amsterdam Model (GLEAM) is one of the remote sensing-based ET and SM products (Martens et al., 2017) comprising a set of algorithms with different inputs of remote sensing observations and meteorological reanalyses. In this study, the monthly GLEAM v3.0 ET product with a spatial resolution of 25 km between 1982 and 2014 was considered. GLEAM ET is provided in mm/month, which is almost equivalent to W/m<sup>2</sup>. In GLEAM, potential ET is computed using the Priestley–Taylor formula based on near-surface temperature and net surface radiation measurements. Second, potential ET is converted to actual ET using a multiple stress factor as a function of remotely sensed soil moisture and vegetation optical depth (Martens et al., 2017).

#### 2.4.3. Fluxnet-MTE

This global LHF product has been produced by upscaling the current Fluxnet-MTE (multitree ensemble) observations of LHF using machine learning techniques and model tree ensembles (Jung et al., 2011). The gridded Fluxnet-MTE data set is derived from continuous in situ measurements of Fluxnet-MTE, remote sensing, and meteorological observations based on model tree ensembles. These data have been extensively validated using observations from eddy covariance systems, and the product has been widely used (Thiery et al., 2015; Y. Zhang et al., 2019; Zeng & Zhang, 2020). These data sets are provided in a monthly time step in MJ/m<sup>2</sup>/d ( $0.0864 \text{ MJ/m}^2/d = 1 \text{ W/m}^2$ ) at a spatial resolution of  $0.5^\circ$  during the 1982–2011 period. It should be noted that a new product was recently released, namely, FLUXCOM (Jung et al., 2019). However, caution was warranted by Jung et al. (2019) when studying trends due to issues with sensor age-based drift in MODIS reflectance. Therefore, this product was not considered in this study.

#### 2.4.4. Climate Research Unit Gridded Time Series (CRU)

We use the updated version of the CRU TS v4 monthly near-surface air temperature data over the period 1901–2014. The spatial resolution of the CRU data sets is 0.5° by 0.5° over the global land surface except for Antarctica. It is derived by the interpolation of monthly climate anomalies using angular-distance weighting from extensive networks of weather station observations (Harris et al., 2020). These data were used to screen out GISS, GLEAM, and Fluxnet-MTE monthly values that correspond to CRU monthly air temperatures lower than 274 K.

#### 2.4.5. Willmott Gridded Time Series (Willmott)

We use the University of Delaware monthly mean surface air temperature data (Willmott et al., 2001) over the period 1900–2014. The spatial resolution of the Willmott data sets is 0.5° by 0.5° over the global land surface. It is derived extensively from the archive of Legates & Willmott (1990) and from the Global Historical Climate Network (GHCN2).





**Figure 2.** Time series of spatially averaged values of all pixels within the irrigation class (50%–100%) of latent heat flux (LHF) (a), SM (b), Pr (c),  $T_{max}$  (d), SW<sub>net</sub> (e), LW<sub>net</sub> (f), and  $R_{net}$  (g) anomalies subtracted from 1900 to 1920 of CMIP6.irr on (blue curve) and CMIP6.noirr (red curve) during the period 1900 to 2014. The shaded area represents  $\pm$  one standard deviation on every side across the Earth system models. Trend values during the 1960–2014 period are added to the panels (\* not significant).

#### 3. Results and Discussion

#### 3.1. CMIP6 Simulations

Historical anomaly time series (relative to a 1900-1920 baseline period) of spatially averaged values of all pixels within the 50%–100% irrigation class for the variables LHF, SM, Pr,  $T_{max}$ ,  $R_{net}$ , LW<sub>net</sub>, and SW<sub>net</sub> are shown in Figure 2. Most of the pixels of this irrigation class are situated into South Asia (see Figure 1a), and we can see that the abrupt change (e.g., LHF and SM) starts from approximately 1960, when a massive irrigation expansion took place as part of the Green Revolution in the 1970s in India (Mishra et al., 2018). Irrigation expansion was enabled by the important rise in the number of groundwater wells (Ambika & Mishra, 2020). In the CMIP6.irr models, the increase in LHF (with a rate of 3.4 W/m<sup>2</sup>/decade) and SM (with a rate of 0.72 mm/decade) resulted in a cooling of the surface (by a rate of -0.16 K/decade) during the 1960-2014 period. This is in line with a recent observational study (Ambika & Mishra, 2020), which demonstrated that irrigation modulates SM and enhances LHF, causing noteworthy cooling in India. This evaporative cooling mechanism was described by Boucher et al. (2004) along with increased absorption of solar radiation. Moreover, Roy et al. (2007) showed a 3-4°C daytime cooling in irrigated areas between the pre-Green Revolution (unirrigated) and post-Green Revolution (irrigated) periods. There is a slight increase in Pr (with a rate of 0.06 mm/day/decade) simulated by CMIP6.irr compared to CMIP6. noirr. The mostly irrigated class mostly corresponds to India, which is a recognized hotspot of land-atmosphere coupling (Koster et al., 2004). The response of rainfall to land conditions is therefore strongly modulated by atmospheric boundary layer and convective parameterizations (Cheruy et al., 2013), but the CMIP6 experimental design does not readily permit assessment of these feedbacks.

Based on SW<sub>net</sub>, there was a solar dimming trend for both irrigated and non-irrigated CMIP6 model types. Cook et al. (2015) related this dimming to irrigation-induced increases in cloud cover. Moreover, the dimming trend over India was recently confirmed using the Global Energy Balance Archive data (GEBA), which is based on ground-based stations measuring energy fluxes (Moseid et al., 2020). The latter study validated NorESM2-LM, CanESM5, MIROC6, CESM2, and CNRM-ESM2-1 SW<sub>net</sub> simulations against the GEBA observations and found similar dimming trends among all models and observations. They found a dimming trend, which is also consistent with several other observational studies (Allen et al., 2013; Singh & Kumar, 2016).





Figure 3. Time series of spatially averaged values of all pixels for all irrigation classes of latent heat flux anomalies subtracted from 1900 to 1920 of CMIP6.irr on (blue curve) and CMIP6.noirr (red curve) during the period 1900 to 2014. Sen's slope per decade over the 1960–2014 period is added to the panels (\* not significant).

This dimming is mainly attributed to an increase in aerosol optical depth (Soni et al., 2016). However, the SW<sub>net</sub> trends in CMIP6.irr and CMIP6.noirr are indistinguishable; therefore, there does not appear to be a link between dimming and irrigation. In contrast, there is an increase in the LW<sub>net</sub> simulated by CMIP6.irr (with a rate of 2.9 W/m<sup>2</sup>/decade), which is reflected in  $R_{net}$  anomalies. The variability (shaded area in Figure 2, which represents  $\pm$  one standard deviation) of CMIP6.irr is larger than that of CMIP6.noirr. Figure 2 is reproduced in Figure S2 in Supporting Information S1 with showing the different values of the three models of CMIP.irr. This variability comes from GISS.E2.1.G as the other two models, which share the same land model and irrigation scheme, tend to agree on each other. Moreover, over this irrigation class, the mean of all variables during 2005–2014 for the three models in CMIP6.irr and the ensemble CMIP6.noirr is shown in Figure S3 in Supporting Information S1. The mean values of LHF and SM of the three models (CMIP6.noirr. While NorESM2-LM has the lowest value for Lwnet, CESM2 has the lowest values for  $R_{net}$  and Swnet. This variability among the CMIP6. irr models may be explained by the different parametrizations of the irrigation implementation.

Time series of spatially averaged values of all pixels within the other irrigation classes (i.e., 0%–50%) for LHF, SM, Pr,  $T_{max}$ ,  $R_{net}$ , LW<sub>net</sub>, and SW<sub>net</sub> anomalies (i.e., subtracted from 1900 to 1920) of CMIP6.irr on and CMIP6. noirr during the period 1900 to 2014 are depicted in Figures S4–S8 in Supporting Information S1. As an example, the behavior of CMIP6.irr and CMIP6.noirr LHF over all irrigation classes is illustrated in Figure 3. Over regions where irrigated area is essentially zero (i.e., 0%-0.1%), the temporal dynamics of time series anomalies are very similar in both CMIP6.irr and CMIP6.noirr, with an increase since circa 1960. Over this class, the trends of all studied variables are consistent between CMIP6.irr and CMIP6.noirr (Figure S4 in Supporting Information S1). Figure 3 shows that LHF starts to diverge between CMIP6.irr and CMIP6.noirr in the least irrigated class, with slightly lower values of LHF in CMIP6.irr than in CMIP6.noirr. The difference in LHF between the two ensembles becomes clear in the third irrigation class (i.e., 5%-10%), and this difference between CMIP6.irr and CMIP6. noirr LHF increases with the fraction of equipped areas for irrigation, which is confirmed for most variables by Figures S4–S8 in Supporting Information S1. This implies that the differences shown in Figure 2 between CMIP6.irr and CMIP6.noirr over the intensively irrigated regions can be attributed to the irrigation processes. It should be noted that the monthly mean of another variable named "daily maximum temperature"  $(T_{daymax})$  showed similar anomalies (Figure S9 in Supporting Information S1) as  $T_{max}$ . In this figure, however, CMIP6.irr includes only one model, GISS.E2.1.G, as the other two models do not provide this variable. Given the similarity of the

9 of 19

# **Earth's Future**





Grid-cell area equipped for irrigation %

**Figure 4.** Spatially averaged Sen's slope trend per decade (calculated between 1960 and 2014) stratified by the percentage of the FAO area equipped by irrigation (%). The dotted lines represent CMIP6 models without irrigation, and the continuous lines represent models including irrigation. CMIP6 variables used: latent heat flux (surface upward latent heat flux), SM (moisture content of soil layer),  $T_{max}$  (annual maximum of monthly air temperature),  $LW_{net}$  (net surface longwave flux), SW<sub>net</sub> (net shortwave flux), and Pr (precipitation flux).

results using  $T_{daymax}$  or  $T_{max}$ , and as most of CMIP6 models do not provide " $T_{daymax}$ ,"  $T_{max}$  will be presented in the subsequent analyses. Figure S10 in Supporting Information S1 reproduces Figure 3 showing the different values of the three models in CMIP6.irr. The three models tend to agree over irrigation classes up to 50% but GISS. E2.1.G, again, shows higher values particularly over the recent years over the irrigation class 50%–100%.

To better understand the behavior of the CMIP6 models individually and since the substantial change of the anomalies shown in Figure 2 started in 1960, Sen's slope was quantified (per pixel) over the 1960–2014 period for the LHF, SM, Pr,  $T_{max}$ ,  $R_{net}$ , LW<sub>net</sub>, and SW<sub>net</sub> variables simulated by all CMIP6 models used in this study. The resulting Sen's slope values were spatially averaged based on the FAO irrigation map classes (see Figure 1a). Over this period, while the values of LHF and SM have increased in CMIP6.irr models (i.e., CESM2, GISS-E2-1-G, and NorESM2-LM), they have decreased or have no trend for all other CMIP6.noirr models (as illustrated in Figure 4). This clearly suggests that irrigation has a nonnegligible effect on the water and surface energy balance over intensively irrigated areas. The largest increases are observed over irrigated grid cells where the fraction irrigated is greater than 50%. Over this irrigation class,  $T_{max}$  decreased for CMIP6.irr (a slight cooling trend), whereas it increased for the majority of the other CMIP6.noirr models. This divergence between CMIP6.irr and CMIP6.noirr over these regions (mostly grid cells located in South Asia) can most likely be attributed to irrigation, as it is one of the major differences between these two categories of simulations.

For the historical evolution of  $R_{net}$ , the effect of irrigation is only detectable over highly irrigated regions. CMIP6. irr simulations begin to diverge from CMIP6.noirr above the irrigation fraction class 20%–50%. There are similar trends of both the CMIP6.irr and CMIP6.noirr models for SW<sub>net</sub>. This indicates that irrigation has a much larger impact on surface turbulent flux partitioning (latent heat) than on surface shortwave radiation (i.e., clouds), which is in line with previous studies (Cook et al., 2015). No clear influence of irrigation on precipitation was detected. Previous studies (Alter et al., 2015; Huber et al., 2014; Qian et al., 2020) have pointed out that irrigation increases precipitation over or downwind of the irrigated regions, but we do not evaluate precipitation trends associated with increasing irrigation in this study, partly because we anticipate that it will be difficult to identify a clear signal within the context of generally more noisy precipitation trends.

# **Earth's Future**





Grid-cell area equipped for irrigation %

Figure 5. Same as Figure 4, but for 11 ensemble members of the CESM2 model. The Histo.noLu simulation is shown as a red line.

Although the representation of irrigation is identical in NorESM2-LM and CESM2, the simulated trends such as the  $T_{\rm max}$  trends are not. This suggests that the impact of irrigation on the simulated climate can differ between models despite identical irrigation implementation. These differences may originate from substantial differences in the model setup, structure and atmospheric parametrizations (de Vrese & Hagemann, 2018). Although CESM2, GISS-E2-1-G, and NorESM2-LM implement an irrigation scheme, GISS.E2.1.G shows the strongest increase in LHF, SM, and LW<sub>net</sub> over all irrigation classes, followed by NorESM2-LM and CESM2. This is not surprising owing to the different quantities of water applied for irrigation, as there are no common data sets of irrigation forcing, defined standards, or implementation protocols for irrigation in the three climate models (see a full description of the irrigations schemes and implementations in the models in Section 2.3.2).

Potentially, the observed trends in CMIP6.irr may be smaller than the internal variability of CMIP6.noirr trends in CMIP6.noirr by internal variability within the individual ensemble members. Under this hypothesis, the positive trend of LHF, for instance, observed over the 1960–2014 period in models with irrigation, may just be one particular realization ("r1i1p1f1") among all the possible ones. Luckily, several CMIP6 models provide multiple-member ensembles. To examine how the modeled trends can differ due to simulated internal climate variability, we assessed the ensemble members of one CMIP6.irr model (CESM2) and one CMIP6.noirr model (IPSL-CM6A-LR). In addition, LUMIP includes specific runs (Histo-noLu) to evaluate individual land-use impacts, including a simulation without irrigation changes (i.e., irrigated area and all other LULCCs held constant at 1850 levels). This run does not isolate irrigation, although at the high-irrigated area level, it is probably mostly an irrigation signal. This enables direct comparison of the same model with irrigation on and off.

Trends over the 1960–2014 period from 11 members of the CESM2 fully coupled historical ensemble are depicted in Figure 5. All coupled historical ensemble members agree well for the LHF,  $T_{\rm max}$ , and SM and disagree with the Histo-noLu simulation. The behavior of CESM2 Histo-noLu is similar to CMIP6.noirr (Figure 4). Similar to CESM2, trends over the 1960–2014 period from nine IPSL-CM6A-LR coupled historical ensemble members are depicted in Figure S11 in Supporting Information S1. Based on the results of IPSL-CM6A-LR and CESM2, trends are consistent across the ensembles for most fields, indicating that the forced trend due to climate change or climate change plus irrigation in the case of CESM2 dominates over the internal variability.





**Figure 6.** (a) Time series of annual latent heat flux (W/m<sup>2</sup>) anomalies (relative to the period 1983–1989) of CMIP6.irr (blue curve) and CMIP6.noirr (red curve) as well as the average of all CMIP6 models (orange curve) during the period 1983 to 2014 along with Fluxnet-MTE, Global Land Surface Satellite (GLASS), and Global Land Evaporation Amsterdam Model (GLEAM). (b) Time series air temperature (K) anomalies (relative to the 1940–1959 period) of CMIP6.irr (blue curve) and CMIP6.noirr (red curve) during the period 1940 to 2014 along with two ground-based air temperature observations: CRU and Willmott over the 50%–100% irrigation class (see Figure 1a). The shaded area represents  $\pm$  one standard deviation of each product.

#### 3.2. Observational Validation

Our results show that the differences in LHF and  $T_{\text{max}}$  between the CMIP6.irr and CMIP6.noirr simulations can be highly linked to irrigation influences. Therefore, to validate whether they make the CMIP6.irr climate models more realistic, we compared the simulated time series of LHF and  $T_{\text{max}}$  to those of observation-based data sets.

We first focused on heavily irrigated regions (i.e., 50%–100% class in Figure 1a) and used three observation-based estimates (Fluxnet-MTE, GLASS, and GLEAM) for LHF, with a common availability period over 1983–2014. The validation period for  $T_{\rm max}$  is longer, and we consider the 1960–2014 period (used in Figure 4) against CRU and Willmott. For both variables, simulated and observed time series are offset to share the same mean over the beginning of the analyzed period for easier trend comparison.

Figure 6a shows the resulting anomalies (relative to 1983–1989) along with those of the averaged CMIP6.irr and CMIP6.noirr models. While the three observation-based data and the CMIP6.irr models showed an increase in LHF, the CMIP6.noirr models showed no increase or decrease but remained steady during the 1983-2014 period. CMIP6.irr  $(3.62 \text{ W/m}^2)$  is closer to the three observations (Fluxnet = 2.38, GLASS = 4.15, and GLEAM = 3.02 W/m<sup>2</sup>) than both CMIP6.all (0.88 W/m<sup>2</sup>) and CMIP6.norir (0.33 W/m<sup>2</sup>) in terms of magnitude. The better match between CMIP6.irr and observations is in line with a recent observational study that showed increasing trends of LHF over western India (Pan et al., 2020). The robust positive LHF trend in intensely irrigated regions obtained across a suite of independent observational products adds confidence to the finding that accounting for irrigation increases the ESM realism in terms of near-surface climate representation in intensely irrigated regions. It should be noted that none of the LHF observations used here to evaluate the CMIP6 models account explicitly for irrigation but that in all cases, the effects of irrigation on LHF are indirectly considered via the observations used to retrieve LHF (e.g., remotely sensed soil moisture and LAI). Furthermore, looking at the time series anomaly of the ensemble mean of all CMIP6 models with and without irrigation (CMIP6.all in Figure 6a), the separation of the CMIP6 models according to irrigation gave better agreement with the observations than taking the ensemble mean of all CMIP6 models. This indicates that, in contrast to previous studies (Flato et al., 2013; Loew et al., 2016), the ensemble mean, over intensively irrigated areas, is not always a better choice. Therefore, caution should be exercised when considering the ensemble mean of all different CMIP6 models.

Figure 6b shows the annual maximum of monthly air temperature time series anomalies (relative to 1940–1959) for CMIP6.irr and CMIP6.noirr along with CRU and Willmott over heavily irrigated regions (i.e., 50%–100% class in Figure 1a). Both Willmott and CRU observations show almost the same variability with relatively small differences. The CMIP6.irr models correctly capture the weak decrease in air temperature that started around 1960 (tend to agree slightly better with Willmott). The entire period is subject to global warming, which becomes clearly discernable in both kinds of simulations after 1980 in the non-irrigated areas (Figure S4 in Supporting Information S1), and in the other irrigation's classes for the CMIP6.noirr simulations only (Figures S5–S8 in Supporting Information S1). The CMIP6.irr simulations, in contrast, show more complex evolutions, with a decrease of  $T_{max}$  between ca. 1950 and 1970, probably caused by irrigation intensification, while the  $T_{max}$  increase afterward suggests that the cooling from irrigation is offset by global warming.

However, despite the variability in the observed  $T_{\text{max}}$ , we can estimate that the latter starts to diverge from the CMIP6.irr simulations around 1980, when the observed  $T_{max}$  is rather stable but over the last decade, showing a strong warming trend. The CMIP6.irr models simulate a lower  $T_{\text{max}}$  than the observations, with a decreasing trend, which is not realistic. The CMIP6.irr simulations are not realistic either, with an excessive  $T_{\text{max}}$  over the last two decades, resulting from a too early warming trend. In highly irrigated areas where the cooling effect of irrigation conflicts anthropogenic warming, our results suggest that the observed  $T_{max}$  could be better captured by climate models with irrigation, but with reduces rates. This analysis is consistent with the fact that GISS-E2-1-G overestimates irrigation demand, and all three models assume unlimited water supply. But given that the simulated LHF is well captured by the CMIP6.irr models in the highly irrigated areas, the mismatch of simulated and observed  $T_{\rm max}$  trends may also be due to biases in the three climate models. It could be also that the irrigated models can be wrong in their magnitude and may overestimate irrigation and the resulting ET increase, thus cool the surface layers more than a with a more realistic irrigation. It is worth mentioning that over 2000–2005, while ET from GLASS increased (in 2001), it decreased from FLUXNET-MTE and GLEAM (in 2002). CMIP6.irr and CMIP6.noirr do not show any particular behavior over this period. Such short terms variations are mostly driven by inter-annual variability but in the case of GLASS data it could be related to the merging of several algorithms and data where MODIS, for instance, started in 2000.

To expand this analysis, Figure 7 shows the 1983–2014 trends for both LHF and  $T_{\rm max}$  over all irrigation classes. Except for the least irrigated class, divergence is detectable starting with irrigation classes 5%–10%: the simulated LHF trends agree better with observational data in CMIP6.irr than in CMIP6.noirr. They remain underestimated in CMIP6.irr, however, for the most highly irrigated class (50%–100%), in which the mean trend of CMIP6.irr simulations (2.57 W/m<sup>2</sup>/decade) very satisfactorily matches the observed ones (GLASS with 2.63, GLEAM with 2.54, and Fluxnet-MTE with 2.24 W/m<sup>2</sup>/decade).

As already mentioned, the analysis of the trends of annual maximum of monthly temperature  $(T_{max})$  over 1983–2014 is complicated by anthropogenic warming. In agreement with global land estimates (Dunn et al., 2020; P. Zhang et al., 2019), it is approximately +0.2 K/decade according to the observation data (CRU and Willmott), based on the least irrigated class (<5%) where the simulated temperature trends are hardly sensitive to irrigation (larger trends for CMIP6.irr, probably owing to large model dispersion, as shown in Figure 4 for  $T_{max}$ ). This positive trend in observed  $T_{max}$  significantly decreases with irrigation intensity, to be less than 0.05 K/decade in the most intensively irrigated class. This response of the warming trend to irrigation intensity is hardly captured by climate models that ignore irrigation (CMIP6.noirr), especially given the large model spread. In the models implementing irrigation (CMIP6.irr), the warming trend is reduced with extensive irrigation, but was excessive compared to observations, that may lead to an unrealistic cooling trend during 1983–2014 in the mostly highly irrigated class.

The reduced warming trends found in CMIP6.irr and observations over highly irrigated areas are consistent with the alleviation of hot extremes by expanding irrigation already found by Thiery et al. (2020) by combining observed temperature and irrigation data with historical simulations with a previous version of CESM2. A likely explanation is that irrigation increases LHF, which alleviates extreme temperatures, a relationship that has been observed in India (Bonfils & Lobell, 2007; Chou et al., 2018) and China (Yang et al., 2020; Zhu et al., 2012), although this cooling may potentially not be solely attributed to irrigation alone (van Oldenborgh et al., 2018). The excessive responses of LHF and  $T_{max}$  to irrigation in the CMIP6.irr models may be related to imperfect implementations of the irrigation practices. Additional detection and attribution research are required to analyze



# **Earth's Future**



**Figure 7.** Spatially averaged Sen's slope trend per decade (calculated between 1983 and 2014 for latent heat flux (LHF) (a) and between 1960 and 2014 for  $T_{max}$  (b)) stratified by the percentage of the FAO area equipped by irrigation (%). CMIP6 variables used: LHF (surface latent heat flux) and  $T_{max}$  (air temperature). The average of all CMIP6.irr (blue curve) and CMIP6.noirr (red curve) as well as five satellite-based LHF (Fluxnet-MTE, Global Land Surface Satellite [GLASS], and Global Land Evaporation Amsterdam Model [GLEAM]) and  $T_{max}$  observations (Willmott and CRU).

how much of the extra cooling in CMIP6.irr is due to irrigation. A proper detection and attribution analysis would require running dedicated factorial experiments with and without transient irrigation extent over the historical period with multiple earth system models." Such type of analysis is beyond the scope of our study. Moreover, the gridded temperature products may underestimate the irrigation effect, as stations are typically not located in irrigated croplands and thus may miss the local cooling influence of irrigation.

# 4. Conclusions

Changes in land use and land cover can alter the biophysical properties of the land surface, including albedo, LAI, soil wetness, and roughness, which in turn influence the exchange of water and energy between the land surface and atmosphere. As already shown by Cook et al. (2015), the expansion of irrigated areas influences the long-term global terrestrial water balance via these biogeophysical effects. The present paper generalizes this seminal study by analyzing historical long-term simulations (1900-2014) of LHF, SM, Pr, T<sub>max</sub>, R<sub>net</sub>, SWnet, and LWnet using fully coupled CMIP6 climate models. Three models among the CMIP6 models account for irrigation activities (i.e., CMIP6.irr), while the remainder omit irrigation (CMIP6.noirr). The striking feature is the clear and visual distinction between the CMIP6.noirr and CMIP6.irr models in heavily irrigated regions. Based on time series anomalies and Sen's trend results, a distinct behavior of the three models was observed, with annual LHF and SM generally increasing over heavily irrigated regions. The trend in the corresponding annual maximum of monthly temperature over the irrigated areas showed a cooling trend. We also find that neglecting irrigation reduces the accuracy of LHF trend estimates over heavily irrigated regions, as the behavior of models with irrigation was consistent and confirmed by observational LHF products. In the intensively irrigated areas, the CMIP6.irr models capture the decrease in air temperature that started in 1960, observed by observational data sets. However, the cooling shown by CMIP6.irr after 2010 was not confirmed by the observational air temperature products. Furthermore, all ensemble members of the two CMIP6 models showed similar trends for all the variables, ruling out the possibility that the difference between CMIP6.irr and CMIP6.noirr is due to natural variability.

Based on the recent CMIP6 experiments (with variables considered in this study) and their comparison to observational products, our results highlight the importance of irrigation as a historical climate forcing over intensively irrigated areas. None of the models considered in this study, however, currently differentiates between irrigation techniques, that is, they all apply one single irrigation technique across the globe (typically adding a calculated amount of extra water to the soil surface). This is an evident shortcoming of these models, which exhibit significant biases relative to observed/reported estimates of irrigation amounts. These models also overlook the reliance of irrigation on groundwater supply in many areas (Siebert et al., 2010), and the potential limitation of irrigation in cases of surface or groundwater shortage. Further improvements to the irrigation schemes and underlying land models are thus needed in the future generation of climate models, to better account for past and future evolutions of water resources, near-surface climate, and also food production over irrigated areas, expected to further expand within the next decades because of climatic and demographic pressures (Puy et al., 2020). Such an effort will need to face the challenge of representing realistic irrigation practices (e.g., timing, source of water, amount of irrigation forcing applied, and method of application) into climate models and comparing these parameterizations across modeling groups.

Large increases, compared to present, in global irrigated agriculture that could double by 2050 (Puy et al., 2020), and are projected in 20100 by two future scenarios (SSP3-7.0 and SSP5-8.5), with a range of values between 3.4 and 4.1 million km<sup>2</sup> (Hurtt et al., 2020). Irrigation areas drive irrigation volumes (Puy et al., 2021), but this relationship will not probably remain the same, as water can become limiting, either because irrigated areas have expanded too much or because climate change decreases the available water resources. Future research could consider CMIP6 projections to assess the impact of future changes in irrigation and its impact on climate variables. Once changes in irrigation are adopted by the majority of models as a standard historical anthropogenic climate forcing, its importance for historical changes in climate can be directly compared to other anthropogenic forcings (e.g., greenhouse gases and aerosols) and other LULCC anthropogenic activities (Sterling et al., 2013) that influence ET and therefore the near-surface climate (e.g., fertilizers, urbanization, etc.). Finally, an important issue is to properly quantify the relative weight of these factors of change in the past to try to project them into the future in a detection and attribution framework.

# **Data Availability Statement**

The data that support the findings of this study are openly available. The Historical Irrigation Dataset version 1.0 can be freely obtained from https://mygeohub.org/publications/8/2. The latest version (5.0) of the FAO "Global Map of Irrigation Areas" (Siebert et al., 2013) is available at http://www.fao.org/aquastat/en/geospatial-information/global-maps-irrigated-areas/latest-version/. The Global Land Evaporation Amsterdam Model data is publicly and freely available, it can be obtained from the SFTP-site sftp://hydras.ugent.be, a password to the SFTP can be received within a few minutes after simply filling an email at https://www.gleam.eu/. The Global Land Surface Satellite evapotranspiration product is available at http://www.glass.umd.edu/GPP/AVHRR/. This global FLUXNET-MTE latent heat flux product is available at https://www.bgc-jena.mpg.de/geodb/projects/ Home.php. The updated version of the CRU TS v4 monthly temperature data is available from https://crudata.uea.ac.uk/cru/data/hrg/. The monthly air temperature data from the University of Delaware can be accessed from https://psl.noaa.gov/data/gridded/data.UDel\_AirT\_Precip.html.

#### Acknowledgments

This research work was funded by the CNES (Centre National d'Etudes Spatiales) through the Science TOSCA (Terre Océan Surfaces Continentales et Atmosphère) program. The authors wish to thank Prof Stefan Siebert and Prof Matti Kummu for the useful information about the global FAO irrigation map. The authors wish to thank Dr. Benjamin I. Cook for the useful information about the GISS model. Agnes Ducharne gratefully acknowledges the Belmont Forum support through a grant by the Agence Nationale de la Recherche (ANR) for the BLUE-GEM project (ANR-21-SOIL-0001).

### References

Abatzoglou, J. T., Abelen, S., Seitz, F., Adler, R. F., Huffman, G. J., Chang, A., et al. (2013). Estimated carbon dioxide emissions from tropical deforestation improved by carbon-density maps. *Remote Sensing of Environment*, 7(1), 89–98. https://doi.org/10.3390/rs6087424

Allen, R. J., Norris, J. R., & Wild, M. (2013). Evaluation of multidecadal variability in CMIP5 surface solar radiation and inferred underestimation of aerosol direct effects over Europe, China, Japan, and India. *Journal of Geophysical Research: Atmospheres*, 118(12), 6311–6336. https://doi.org/10.1002/jgrd.50426

Alter, R. E., Fan, Y., Lintner, B. R., & Weaver, C. P. (2015). Observational evidence that Great Plains irrigation has enhanced summer precipitation intensity and totals in the Midwestern United States. *Journal of Hydrometeorology*, 16(4), 1717–1735. https://doi.org/10.1175/ JHM-D-14-0115.1

Al-Yaari, A., Ducharne, A., Cheruy, F., Crow, W. T., & Wigneron, J. P. (2019). Satellite-based soil moisture provides missing link between summertime precipitation and surface temperature biases in CMIP5 simulations over conterminous United States. *Scientific Reports*, 9(1), 1657. https://doi.org/10.1038/s41598-018-38309-5

Ambika, A. K., & Mishra, V. (2020). Substantial decline in atmospheric aridity due to irrigation in India. *Environmental Research Letters*, 15(12), 124060. https://doi.org/10.1088/1748-9326/abc8bc

- Berg, A., Lintner, B. R., Findell, K., Seneviratne, S. I., van den Hurk, B., Ducharne, A., et al. (2014). Interannual coupling between summertime surface temperature and precipitation over land: Processes and implications for climate change. *Journal of Climate*, 28(3), 1308–1328. https:// doi.org/10.1175/jcli-d-14-00324.1
- Bonfils, C., & Lobell, D. (2007). Empirical evidence for a recent slowdown in irrigation-induced cooling. Proceedings of the National Academy of Sciences, 104(34), 13582–13587. https://doi.org/10.1073/pnas.0700144104
- Boucher, O., Myhre, G., & Myhre, A. (2004). Direct human influence of irrigation on atmospheric water vapour and climate. *Climate Dynamics*, 22(6–7), 597–603. https://doi.org/10.1007/s00382-004-0402-4
- Boysen, L., Brovkin, V., Pongratz, J., Lawrence, D., Lawrence, P., Vuichard, N., et al. (2020). Global climate response to idealized deforestation in CMIP6 models. *Biogeosciences Discussions*. https://doi.org/10.5194/bg-2020-229
- Chandler, R. E., & Scott, E. M. (2011). Statistical methods for trend detection and analysis in the environmental sciences. In *Statistical methods for trend detection and analysis in the environmental sciences*. https://doi.org/10.1002/9781119991571
- Chen, L., & Dirmeyer, P. A. (2020). Distinct impacts of land use and land management on summer temperatures. *Frontiers in Earth Science*, 8, 245. https://doi.org/10.3389/feart.2020.00245
- Cheruy, F., Campoy, A., Dupont, J. C., Ducharne, A., Hourdin, F., Haeffelin, M., et al. (2013). Combined influence of atmospheric physics and soil hydrology on the simulated meteorology at the SIRTA atmospheric observatory. *Climate Dynamics*, 40(9), 2251–2269. https://doi. org/10.1007/s00382-012-1469-y
- Cheruy, F., Ducharne, A., Hourdin, F., Musat, I., Vignon, É., Gastineau, G., et al. (2020). Improved near-surface continental climate in IPSL-CM6A-LR by combined evolutions of atmospheric and land surface physics. *Journal of Advances in Modeling Earth Systems*, *12*(10), e2019MS002005. https://doi.org/10.1029/2019MS002005
- Cheruy, F., Dufresne, J. L., Hourdin, F., & Ducharne, A. (2014). Role of clouds and land-atmosphere coupling in midlatitude continental summer warm biases and climate change amplification in CMIP5 simulations. *Geophysical Research Letters*, *41*(18), 6493–6500. https://doi.org/10.1002/2014gl061145
- Chou, C., Ryu, D., Lo, M. H., Wey, H. W., & Malano, H. M. (2018). Irrigation-induced land-atmosphere feedbacks and their impacts on Indian summer monsoon. *Journal of Climate*, 31(21), 8785–8801. https://doi.org/10.1175/JCLI-D-17-0762.1
- Cook, B. I., Mankin, J. S., Marvel, K., Williams, A. P., Smerdon, J. E., & Anchukaitis, K. J. (2020). Twenty-first century drought projections in the CMIP6 forcing scenarios. *Earth's Future*, 8(6), e2019EF001461. https://doi.org/10.1029/2019EF001461
- Cook, B. I., McDermid, S. S., Puma, M. J., Williams, A. P., Seager, R., Kelley, M., et al. (2020). Divergent regional climate Consequences of maintaining current irrigation rates in the 21st century. *Journal of Geophysical Research: Atmospheres*, 125(14), e2019JD031814. https://doi. org/10.1029/2019JD031814
- Cook, B. I., Shukla, S. P., Puma, M. J., & Nazarenko, L. S. (2015). Irrigation as an historical climate forcing. *Climate Dynamics*, 44(5), 1715– 1730. https://doi.org/10.1007/s00382-014-2204-7
- Danabasoglu, G., Lamarque, J. F., Bacmeister, J., Bailey, D. A., DuVivier, A. K., Edwards, J., et al. (2020). The community Earth system model version 2 (CESM2). Journal of Advances in Modeling Earth Systems, 12, e2019MS001916. https://doi.org/10.1029/2019MS001916
- DeAngelis, A., Dominguez, F., Fan, Y., Robock, A., Kustu, M. D., & Robinson, D. (2010). Evidence of enhanced precipitation due to irrigation over the Great Plains of the United States. *Journal of Geophysical Research*, 115(D15), D15115. https://doi.org/10.1029/2010jd013892
- de Rosnay, P., Polcher, J., Laval, K., & Sabre, M. (2003). Integrated parameterization of irrigation in the land surface model ORCHIDEE. Validation over Indian Peninsula. Geophysical Research Letters, 30(19), 1986. https://doi.org/10.1029/2003g1018024
- de Vrese, P., & Hagemann, S. (2018). Uncertainties in modelling the climate impact of irrigation. Climate Dynamics, 51(5–6), 2023–2038. https:// doi.org/10.1007/s00382-017-3996-z
- de Vrese, P., & Stacke, T. (2020). Irrigation and hydrometeorological extremes. Climate Dynamics, 55(5–6), 1521–1537. https://doi.org/10.1007/ s00382-020-05337-9
- Dunn, R. J. H., Alexander, L. V., Donat, M. G., Zhang, X., Bador, M., Herold, N., et al. (2020). Development of an updated global land in situ-based data set of temperature and precipitation extremes: HadEX3. *Journal of Geophysical Research: Atmospheres*, 125(16), e2019JD032263. https://doi.org/10.1029/2019JD032263
- Eyring, V., Bony, S., Meehl, G. A., Senior, C. A., Stevens, B., Stouffer, R. J., & Taylor, K. E. (2016). Overview of the Coupled Model Intercomparison Project Phase 6 (CMIP6) experimental design and organization. *Geoscientific Model Development*, 9(5), 1937–1958. https://doi. org/10.5194/gmd-9-1937-2016
- Felfelani, F., Lawrence, D. M., & Pokhrel, Y. (2021). Representing intercell lateral groundwater flow and aquifer pumping in the Community Land Model. Water Resources Research, 57(1), e2020WR027531. https://doi.org/10.1029/2020WR027531
- Flato, G., Marotzke, J., Abiodun, B., Braconnot, P., Chou, S. C., Collins, W., et al. (2013). Evaluation of climate models. In Climate Change 2013: The Physical Science Basis. Contribution of Working Group I to the Fifth Assessment Report of the Intergovernmental Panel on Climate Change. Cambridge University Press.
- Foley, J. A., Ramankutty, N., Brauman, K. A., Cassidy, E. S., Gerber, J. S., Johnston, M., et al. (2011). Solutions for a cultivated planet. *Nature*, 478(7369), 337–342. https://doi.org/10.1038/nature10452
- Goldewijk, K. K., Beusen, A., Doelman, J., & Stehfest, E. (2017). Anthropogenic land use estimates for the Holocene HYDE 3.2. Earth System Science Data, 9(2), 927–953. https://doi.org/10.5194/essd-9-927-2017
- Gormley-Gallagher, A. M., Sterl, S., Hirsch, A. L., Seneviratne, S. I., Davin, E. L., & Thiery, W. (2020). Agricultural management effects on mean and extreme temperature trends. *Earth System Dynamics*, 1–27. https://doi.org/10.5194/esd-2020-35
- Guimberteau, M., Laval, K., Perrier, A., & Polcher, J. (2012). Global effect of irrigation and its impact on the onset of the Indian summer monsoon. *Climate Dynamics*, 39(6), 1329–1348. https://doi.org/10.1007/s00382-011-1252-5
- Haddeland, I., Lettenmaier, D. P., & Skaugen, T. (2006). Effects of irrigation on the water and energy balances of the Colorado and Mekong river basins. *Journal of Hydrology*, 324(1–4), 210–223. https://doi.org/10.1016/j.jhydrol.2005.09.028
- Hajima, T., Watanabe, M., Yamamoto, A., Tatebe, H., Noguchi, M. A., Abe, M., et al. (2020). Development of the MIROC-ES2L Earth system model and the evaluation of biogeochemical processes and feedbacks. Geoscientific Model Development. https://doi.org/10.5194/gmd-13-2197-2020
- Harris, I., Osborn, T. J., Jones, P., & Lister, D. (2020). Version 4 of the CRU TS monthly high-resolution gridded multivariate climate dataset. Scientific Data, 7(1), 109. https://doi.org/10.1038/s41597-020-0453-3
- Hauser, M., Thiery, W., & Isabelle Seneviratne, S. (2019). Potential of global land water recycling to mitigate local temperature extremes. *Earth System Dynamics*, 10(1), 157–169. https://doi.org/10.5194/esd-10-157-2019
- Hirsch, A. L., Prestele, R., Davin, E. L., Seneviratne, S. I., Thiery, W., & Verburg, P. H. (2018). Modelled biophysical impacts of conservation agriculture on local climates. *Global Change Biology*, 24(10), 4758–4774. https://doi.org/10.1111/gcb.14362

Hirsch, A. L., Wilhelm, M., Davin, E. L., Thiery, W., & Seneviratne, S. I. (2017). Can climate-effective land management reduce regional warming? Journal of Geophysical Research: Atmospheres, 122(4), 2269–2288. https://doi.org/10.1002/2016JD026125

Huber, D. B., Mechem, D. B., & Brunsell, N. A. (2014). The effects of Great Plains irrigation on the surface energy balance, regional circulation, and precipitation. *Climate*, 2(2), 103–128. https://doi.org/10.3390/cli2020103

Hurtt, G. C., Chini, L., Sahajpal, R., Frolking, S., Bodirsky, B. L., Calvin, K., et al. (2020). Harmonization of global land use change and management for the period 850–2100 (LUH2) for CMIP6. Geoscientific Model Development. https://doi.org/10.5194/gmd-13-5425-2020

- Jha, R., Mondal, A., Devanand, A., Roxy, M. K., & Ghosh, S. (2022). Limited influence of irrigation on pre-monsoon heat stress in the Indo-Gangetic Plain. *Nature Communications*, 13(1), 4275. https://doi.org/10.1038/s41467-022-31962-5
- Jiang, C., Ryu, Y., Fang, H., Myneni, R., Claverie, M., & Zhu, Z. (2017). Inconsistencies of interannual variability and trends in long-term satellite leaf area index products. *Global Change Biology*, 23(10), 4133–4146. https://doi.org/10.1111/gcb.13787
- Jung, M., Koirala, S., Weber, U., Ichii, K., Gans, F., Camps-Valls, G., et al. (2019). The FLUXCOM ensemble of global land-atmosphere energy fluxes. *Scientific Data*, 6(1), 74. https://doi.org/10.1038/s41597-019-0076-8
- Jung, M., Reichstein, M., Margolis, H. A., Cescatti, A., Richardson, A. D., Arain, M. A., et al. (2011). Global patterns of land-atmosphere fluxes of carbon dioxide, latent heat, and sensible heat derived from eddy covariance, satellite, and meteorological observations. *Journal of Geophysical Research*, 116, G00J07. https://doi.org/10.1029/2010JG001566
- Kanamaru, H., & Kanamitsu, M. (2008). Model diagnosis of nighttime minimum temperature warming during summer due to irrigation in the California Central Valley. *Journal of Hydrometeorology*, 9(5), 1061–1072. https://doi.org/10.1175/2008JHM967.1
- Kelley, M., Schmidt, G. A., Nazarenko, L. S., Bauer, S. E., Ruedy, R., Russell, G. L., et al. (2020). GISS-E2.1: Configurations and climatology. Journal of Advances in Modeling Earth Systems, 12(8), e2019MS002025. https://doi.org/10.1029/2019MS002025

Kendall, M. G. (1957). Rank correlation methods (4th ed.). Charles Griffin.

- Kim, H., & Lakshmi, V. (2019). Global dynamics of stored precipitation water in the topsoil layer from satellite and reanalysis data. Water Resources Research, 55(4), 3328–3346. https://doi.org/10.1029/2018WR023166
- Kim, Y., Moorcroft, P. R., Aleinov, I., Puma, M. J., & Kiang, N. Y. (2015). Variability of phenology and fluxes of water and carbon with observed and simulated soil moisture in the Ent Terrestrial Biosphere Model (Ent TBM version 1.0.1.0.0). *Geoscientific Model Development*, 8(12), 3837–3865. https://doi.org/10.5194/gmd-8-3837-2015
- Koster, R. D., Suarez, M. J., Liu, P., Jambor, U., Berg, A., Kistler, M., et al. (2004). Realistic initialization of land surface states: Impacts on subseasonal forecast skill. *Journal of Hydrometeorology*, 5(6), 1049–1063. https://doi.org/10.1175/jhm-387.1
- Lahoz, W. A., & de Lannoy, G. J. M. (2014). Closing the gaps in our knowledge of the hydrological cycle over land: Conceptual problems. Surveys in Geophysics, 35(3), 623–660. https://doi.org/10.1007/s10712-013-9221-7
- Lawrence, D. M., Fisher, R., Koven, C., Oleson, K., Swenson, S., & Vertenstein, M. (2018). Technical description of version 5.0 of the Community Land Model (CLM). NCAR/TN-478+STR NCAR Technical Note.
- Lawrence, D. M., Fisher, R. A., Koven, C. D., Oleson, K. W., Swenson, S. C., Bonan, G., et al. (2019). The Community Land Model version 5: Description of new features, benchmarking, and impact of forcing uncertainty. *Journal of Advances in Modeling Earth Systems*, 11(12), 4245–4287. https://doi.org/10.1029/2018MS001583
- Lawrence, D. M., Hurtt, G. C., Arneth, A., Brovkin, V., Calvin, K. V., Jones, A. D., et al. (2016). The land use model intercomparison project (LUMIP) contribution to CMIP6: Rationale and experimental design. *Geoscientific Model Development*, 9(9), 2973–2998. https://doi. org/10.5194/gmd-9-2973-2016
- Lawston, P. M., Santanello, J. A., Zaitchik, B. F., & Rodell, M. (2015). Impact of irrigation methods on land surface model spinup and initialization of WRF forecasts. *Journal of Hydrometeorology*, 16(3), 1135–1154. https://doi.org/10.1175/JHM-D-14-0203.1
- Legates, D. R., & Willmott, C. J. (1990). Mean seasonal and spatial variability in global surface air temperature. *Theoretical and Applied Climatology*, 41(1–2), 11–21. https://doi.org/10.1007/BF00866198
- Loew, A., Andersson, A., Trentmann, J., & Schröder, M. (2016). Assessing surface solar radiation fluxes in the CMIP ensembles. Journal of Climate, 29(20), 7231–7246. https://doi.org/10.1175/JCLI-D-14-00503.1
- Lombardozzi, D. L., Lu, Y., Lawrence, P. J., Lawrence, D. M., Swenson, S., Oleson, K. W., et al. (2020). Simulating agriculture in the Community Land Model version 5. Journal of Geophysical Research: Biogeosciences, 125(8), e2019JG005529. https://doi.org/10.1029/2019JG005529
- Lu, Y., Harding, K., & Kueppers, L. (2017). Irrigation effects on land-atmosphere coupling strength in the United States. *Journal of Climate*, 30(10), 3671–3685. https://doi.org/10.1175/jcli-d-15-0706.1
- Lu, Y., Jin, J., & Kueppers, L. M. (2015). Crop growth and irrigation interact to influence surface fluxes in a regional climate-cropland model (WRF3.3-CLM4crop). Climate Dynamics, 45(11–12), 3347–3363. https://doi.org/10.1007/s00382-015-2543-z
- Luyssaert, S., Jammet, M., Stoy, P. C., Estel, S., Pongratz, J., Ceschia, E., et al. (2014). Land management and land-cover change have impacts of similar magnitude on surface temperature. *Nature Climate Change*, 4(5), 389–393. https://doi.org/10.1038/nclimate2196
- Mahmood, R., Pielke, R. A., Hubbard, K. G., Niyogi, D., Dirmeyer, P. A., Mcalpine, C., et al. (2014). Land cover changes and their biogeophysical effects on climate. *International Journal of Climatology*, 34(4), 929–953. https://doi.org/10.1002/joc.3736
- Martens, B., Miralles, D. G., Lievens, H., Van Der Schalie, R., De Jeu, R. A. M., Fernández-Prieto, D., et al. (2017). GLEAM v3: Satellite-based land evaporation and root-zone soil moisture. *Geoscientific Model Development*, 10(5), 1903–1925. https://doi.org/10.5194/gmd-10-1903-2017
- Meier, J., Zabel, F., & Mauser, W. (2018). A global approach to estimate irrigated areas A comparison between different data and statistics. *Hydrology and Earth System Sciences*, 22(2), 1119–1133. https://doi.org/10.5194/hess-22-1119-2018
- Miller, R. L., Schmidt, G. A., Nazarenko, L. S., Bauer, S. E., Kelley, M., Ruedy, R., et al. (2021). CMIP6 historical simulations (1850–2014) with GISS-e2.1. Journal of Advances in Modeling Earth Systems, 13(1), e2019MS002034. https://doi.org/10.1029/2019MS002034
- Milly, P. C. D., Wetherald, R. T., Dunne, K. A., & Delworth, T. L. (2002). Increasing risk of great floods in a changing climate. *Nature*, 415(6871), 514–517. https://doi.org/10.1038/415514a
- Mishra, V., Shah, R., Azhar, S., Shah, H., Modi, P., & Kumar, R. (2018). Reconstruction of droughts in India using multiple land-surface models (1951–2015). *Hydrology and Earth System Sciences*, 22(4), 2269–2284. https://doi.org/10.5194/hess-22-2269-2018
- Mizuochi, H., Ducharne, A., Cheruy, F., Ghattas, J., Al-Yaari, A., Wigneron, J.-P., et al. (2021). Multi-variable evaluation of land surface processes in forced and coupled modes reveals new error sources to the simulated water cycle in the IPSL climate model. *Hydrology and Earth* System Sciences Discussions, 25(4), 2199–2221. https://doi.org/10.5194/hess-25-2199-2021
- Moseid, K. O., Schulz, M., Storelvmo, T., Julsrud, I. R., Olivié, D., Nabat, P., et al. (2020). Bias in CMIP6 models as compared to observed regional dimming and brightening. *Atmospheric Chemistry and Physics*, 20, 16023–16040. https://doi.org/10.5194/acp-20-16023-2020
- Müller, W. A., Jungclaus, J. H., Mauritsen, T., Baehr, J., Bittner, M., Budich, R., et al. (2018). A higher-resolution version of the Max Planck Institute Earth system model (MPI-ESM1.2-HR). *Journal of Advances in Modeling Earth Systems*, 10(7), 1383–1413. https://doi.org/10.1029/2017MS001217



- Oki, T. (1999). The global water cycle. In K. Browning & R. Gurney (Eds.), *Global energy and water cycles* (pp. 10–27). Cambridge University Press.
- Ozdogan, M., Yang, Y., Allez, G., & Cervantes, C. (2010). Remote sensing of irrigated agriculture: Opportunities and challenges. *Remote Sensing*, 2(9), 2274–2304. https://doi.org/10.3390/rs2092274
- Pan, S., Pan, N., Tian, H., Friedlingstein, P., Sitch, S., Shi, H., et al. (2020). Evaluation of global terrestrial evapotranspiration using state-of-theart approaches in remote sensing, machine learning and land surface modeling. *Hydrology and Earth System Sciences*, 24(3), 1485–1509. https://doi.org/10.5194/hess-24-1485-2020
- Pitman, A. J., De Noblet-Ducoudré, N., Avila, F. B., Alexander, L. V., Boisier, J. P., Brovkin, V., et al. (2012). Effects of land cover change on temperature and rainfall extremes in multi-model ensemble simulations. *Earth System Dynamics*, 3(2), 213–231. https://doi.org/10.5194/ esd-3-213-2012
- Pokhrel, Y., Hanasaki, N., Koirala, S., Cho, J., Yeh, P. J. F. F., Kim, H., et al. (2011). Incorporating anthropogenic water regulation modules into a land surface model. *Journal of Hydrometeorology*, 13(1), 255–269. https://doi.org/10.1175/jhm-d-11-013.1
- Pokhrel, Y., Hanasaki, N., Wada, Y., & Kim, H. (2016). Recent progresses in incorporating human land-water management into global land surface models toward their integration into Earth system models. Wiley Interdisciplinary Reviews: Water, 3(4), 548–574. https://doi. org/10.1002/wat2.1150
- Prentice, I. C., Meng, T., Wang, H., Harrison, S. P., Ni, J., & Wang, G. (2011). Evidence of a universal scaling relationship for leaf CO<sub>2</sub> drawdown along an aridity gradient. *New Phytologist*, 190(1), 169–180. https://doi.org/10.1111/j.1469-8137.2010.03579.x
- Pryor, S. C., Sullivan, R. C., & Wright, T. (2016). Quantifying the roles of changing Albedo, emissivity, and energy partitioning in the impact of irrigation on atmospheric heat content. *Journal of Applied Meteorology and Climatology*, 55(8), 1699–1706. https://doi.org/10.1175/ JAMC-D-15-0291.1
- Puma, M. J., & Cook, B. I. (2010). Effects of irrigation on global climate during the 20th century. Journal of Geophysical Research, 115(D16), D16120. https://doi.org/10.1029/2010JD014122

Puy, A. (2018). Irrigated areas grow faster than the population. Ecological Applications, 28(6), 1413–1419. https://doi.org/10.1002/eap.1743

- Puy, A., Borgonovo, E., lo Piano, S., Levin, S. A., & Saltelli, A. (2021). Irrigated areas drive irrigation water withdrawals. *Nature Communica*tions, 12(1), 4525. https://doi.org/10.1038/s41467-021-24508-8
- Puy, A., Lo Piano, S., & Saltelli, A. (2020). Current models underestimate future irrigated areas. *Geophysical Research Letters*, 47(8), e2020GL087360. https://doi.org/10.1029/2020GL087360
- Qian, Y., Yang, Z., Feng, Z., Liu, Y., Gustafson, W. I., Berg, L. K., et al. (2020). Neglecting irrigation contributes to the simulated summertime warm-and-dry bias in the central United States. Npj Climate and Atmospheric Science, 3(1), 31. https://doi.org/10.1038/s41612-020-00135-w
- Rost, S., Gerten, D., Bondeau, A., Lucht, W., Rohwer, J., & Schaphoff, S. (2008). Agricultural green and blue water consumption and its influence on the global water system. *Water Resources Research*, 44(9), W09405. https://doi.org/10.1029/2007WR006331
- Roy, S. S., Mahmood, R., Niyogi, D., Lei, M., Foster, S. A., Hubbard, K. G., et al. (2007). Impacts of the agricultural Green Revolution-induced land use changes on air temperatures in India. *Journal of Geophysical Research*, 112(21), D21108. https://doi.org/10.1029/2007JD008834
- Sacks, W. J., Cook, B. I., Buenning, N., Levis, S., & Helkowski, J. H. (2009). Effects of global irrigation on the near-surface climate. *Climate Dynamics*, 33(2–3), 159–175. https://doi.org/10.1007/s00382-008-0445-z
- Saeed, F., Hagemann, S., & Jacob, D. (2009). Impact of irrigation on the South Asian summer monsoon. *Geophysical Research Letters*, 36(20), L20711. https://doi.org/10.1029/2009GL040625
- Seland, Ø., Bentsen, M., Olivié, D., Toniazzo, T., Gjermundsen, A., Graff, L. S., et al. (2020). Overview of the Norwegian Earth System Model (NorESM2) and key climate response of CMIP6 DECK, historical, and scenario simulations. *Geoscientific Model Development*, 13, 6165– 6200. https://doi.org/10.5194/gmd-13-6165-2020
- Sellers, P. J., Dickinson, R. E., Randall, D. A., Betts, A. K., Hall, F. G., Berry, J. A., et al. (1997). Modeling the exchanges of energy, water, and carbon between continents and the atmosphere. *Science*, 275(5299), 502–509. https://doi.org/10.1126/science.275.5299.502
- Sen, P. K. (1968). Estimates of the regression coefficient based on Kendall's tau. *Journal of the American Statistical Association*, 63(324), 1379–1389. https://doi.org/10.1080/01621459.1968.10480934
- Shukla, S. P., Puma, M. J., & Cook, B. I. (2014). The response of the South Asian Summer Monsoon circulation to intensified irrigation in global climate model simulations. *Climate Dynamics*, 42(1–2), 21–36. https://doi.org/10.1007/s00382-013-1786-9
- Siebert, S., Burke, J., Faures, J. M., Frenken, K., Hoogeveen, J., Döll, P., & Portmann, F. T. (2010). Groundwater use for irrigation A global inventory. *Hydrology and Earth System Sciences*, 14(10), 1863–1880. https://doi.org/10.5194/hess-14-1863-2010
- Siebert, S., Henrich, V., Frenken, K., & Burke, J. (2013). Update of the digital global map of irrigation areas to version 5. In *Rheinische Friedrich-Wilhelms-Universituy, Bonn, Germany and Food and Agriculture Organization of the United Nations, Rome, Italy.* https://doi.org/10.13140/2.1.2660.6728
- Siebert, S., Kummu, M., Porkka, M., Döll, P., Ramankutty, N., & Scanlon, B. R. (2015). A global data set of the extent of irrigated land from 1900 to 2005. *Hydrology and Earth System Sciences*, 19(3), 1521–1545. https://doi.org/10.5194/hess-19-1521-2015
- Singh, J., & Kumar, M. (2016). Solar radiation over four cities of India: Trend analysis using Mann-Kendall Test. International Journal of Renewable Energy Resources, 6, 1385–1395. https://doi.org/10.1234/ijrer.v6i4.4572
- Soni, V. K., Pandithurai, G., & Pai, D. S. (2016). Is there a transition of solar radiation from dimming to brightening over India? Atmospheric Research, 169, 209–224. https://doi.org/10.1016/j.atmosres.2015.10.010
- Sorooshian, S., AghaKouchak, A., & Li, J. (2014). Influence of irrigation on land hydrological processes over California. Journal of Geophysical Research: Atmospheres, 119(23), 13137–13152. https://doi.org/10.1002/2014JD022232
- Stanfield, R. E., Jiang, J. H., Dong, X., Xi, B., Su, H., Donner, L., et al. (2016). A quantitative assessment of precipitation associated with the ITCZ in the CMIP5 GCM simulations. *Climate Dynamics*, 47(5–6), 1863–1880. https://doi.org/10.1007/s00382-015-2937-y
- Sterling, S. M., Ducharne, A., & Polcher, J. (2013). The impact of global land-cover change on the terrestrial water cycle. *Nature Climate Change*, 3(4), 385–390. https://doi.org/10.1038/nclimate1690
- Swart, N. C., Cole, J. N. S., Kharin, V. V., Lazare, M., Scinocca, J. F., Gillett, N. P., et al. (2019). The Canadian Earth System Model version 5 (CanESM5.0.3). Geoscientific Model Development, 12(11), 4823–4873. https://doi.org/10.5194/gmd-12-4823-2019
- Tabari, H., Marofi, S., Aeini, A., Talaee, P. H., & Mohammadi, K. (2011). Trend analysis of reference evapotranspiration in the western half of Iran. Agricultural and Forest Meteorology, 151(2), 128–136. https://doi.org/10.1016/j.agrformet.2010.09.009
- Thiery, W., Davin, E. L., Lawrence, D. M., Hirsch, A. L., Hauser, M., & Seneviratne, S. I. (2017). Present-day irrigation mitigates heat extremes. Journal of Geophysical Research: Atmospheres, 122(3), 1403–1422. https://doi.org/10.1002/2016JD025740
- Thiery, W., Davin, E. L., Panitz, H. J., Demuzere, M., Lhermitte, S., & Van Lipzig, N. (2015). The impact of the African Great Lakes on the regional climate. *Journal of Climate*, 28(10), 4061–4085. https://doi.org/10.1175/JCLI-D-14-00565.1

- 23284277, 2022, 11, Downloaded from doi/10.1029/2022EF002859 by Universitä Library on [22/11/2022]. See the Terms and Condit ;OA
- Thiery, W., Visser, A. J., Fischer, E. M., Hauser, M., Hirsch, A. L., Lawrence, D. M., et al. (2020). Warming of hot extremes alleviated by expanding irrigation. *Nature Communications*, 11(1), 290. https://doi.org/10.1038/s41467-019-14075-4
- Vanderkelen, I., van Lipzig, N. P. M., Sacks, W. J., Lawrence, D. M., Clark, M. P., Mizukami, N., et al. (2021). Simulating the impact of global reservoir expansion on the present-day climate. *Journal of Geophysical Research: Atmospheres*, 126(16), e2020JD034485. https://doi. org/10.1029/2020JD034485
- van Oldenborgh, G. J., Philip, S., Kew, S., van Weele, M., Uhe, P., Otto, F., et al. (2018). Extreme heat in India and anthropogenic climate change. Natural Hazards and Earth System Sciences, 18(1), 365–381. https://doi.org/10.5194/nhess-18-365-2018
- Vicente-Serrano, S. M., Peña-Gallardo, M., Hannaford, J., Murphy, C., Lorenzo-Lacruz, J., Dominguez-Castro, F., et al. (2019). Climate, irrigation, and land cover change explain streamflow trends in Countries bordering the Northeast Atlantic. *Geophysical Research Letters*, 46(19), 10821–10833. https://doi.org/10.1029/2019GL084084
- Vörösmarty, C. J., Sahagian, D., Vorosmarty, C. J., & Sahagian, D. (2000). Anthropogenic disturbance of the terrestrial water cycle. *BioScience*, 50(9), 753. https://doi.org/10.1641/0006-3568(2000)050[0753:ADOTTW]2.0.CO;2
- Willmott, C. J., Matsuura, K., & Legates, D. R. (2001). Terrestrial air temperature and precipitation: Monthly and annual time series (1950–1999). Center for Climate Research Version, 1, 179–186. https://doi.org/10.3354/cr001179
- Wisser, D., Fekete, B. M., Vörösmarty, C. J., & Schumann, A. H. (2010). Reconstructing 20th century global hydrography: A contribution to the global terrestrial network-hydrology (GTN-H). Hydrology and Earth System Sciences, 14(1), 1–24. https://doi.org/10.5194/hess-14-1-2010
- Wu, T., Lu, Y., Fang, Y., Xin, X., Li, L., Li, W., et al. (2019). The Beijing Climate Center Climate System Model (BCC-CSM): The main progress from CMIP5 to CMIP6. Geoscientific Model Development. https://doi.org/10.5194/gmd-12-1573-2019
- Yang, Q., Huang, X., & Tang, Q. (2020). Irrigation cooling effect on land surface temperature across China based on satellite observations. Science of the Total Environment, 705, 135984. https://doi.org/10.1016/j.scitotenv.2019.135984
- Yao, Y., Liang, S., Li, X., Hong, Y., Fisher, J. B., Zhang, N., et al. (2014). Bayesian multimodel estimation of global terrestrial latent heat flux from eddy covariance, meteorological, and satellite observations. *Journal of Geophysical Research: Atmospheres*, 119(8), 4521–4545. https:// doi.org/10.1002/2013JD020864
- Zeng, J., & Zhang, Q. (2020). The trends in land surface heat fluxes over global monsoon domains and their responses to monsoon and precipitation. *Scientific Reports*, 10(1), 5762. https://doi.org/10.1038/s41598-020-62467-0
- Zhang, P., Ren, G., Xu, Y., Wang, X. L., Qin, Y., Sun, X., & Ren, Y. (2019). Observed changes in extreme temperature over the global land based on a newly developed station daily dataset. *Journal of Climate*, 32(24), 8489–8509. https://doi.org/10.1175/JCLI-D-18-0733.1
- Zhang, Y., Zhou, S., Gentine, P., & Xiao, X. (2019). Can vegetation optical depth reflect changes in leaf water potential during soil moisture dry-down events? *Remote Sensing of Environment*, 234, 111451. https://doi.org/10.1016/j.rse.2019.111451
- Zhu, X., Liang, S., & Pan, Y. Z. (2012). Observational evidence of the cooling effect of agricultural irrigation in Jilin, China. *Climatic Change*, 114(3-4), 799–811. https://doi.org/10.1007/s10584-012-0435-3

## **References From the Supporting Information**

- Adcroft, A., Anderson, W., Balaji, V., Blanton, C., Bushuk, M., Dufour, C. O., et al. (2019). The GFDL global ocean and sea ice model OM4.0: Model description and simulation features. *Journal of Advances in Modeling Earth Systems*, 11(10), 3167–3211. https://doi. org/10.1029/2019MS001726
- Bi, D., Dix, M., Marsland, S., O'farrell, S., Sullivan, A., Bodman, R., et al. (2020). Configuration and spin-up of ACCESS-CM2, the new generation Australian Community climate and Earth system simulator coupled model. *Journal of Southern Hemisphere Earth Systems Science*, 70(1), 225. https://doi.org/10.1071/ES19040
- Boucher, O., Servonnat, J., Albright, A. L., Aumont, O., Balkanski, Y., Bastrikov, V., et al. (2020). Presentation and evaluation of the IPSL-CM6A-LR climate model. *Journal of Advances in Modeling Earth Systems*, 12(7), e2019MS002010. https://doi.org/10.1029/2019MS002010
- Golaz, J. C., Caldwell, P. M., Van Roekel, L. P., Petersen, M. R., Tang, Q., Wolfe, J. D., et al. (2019). The DOE E3SM coupled model version 1: Overview and evaluation at standard resolution. *Journal of Advances in Modeling Earth Systems*, 11(7), 2089–2129. https://doi. org/10.1029/2018MS001603
- Gutjahr, O., Putrasahan, D., Lohmann, K., Jungclaus, J. H., Von Storch, J. S., Brüggemann, N., et al. (2019). Max Planck Institute Earth System Model (MPI-ESM1.2) for the high-resolution model intercomparison project (HighResMIP). Geoscientific Model Development. https://doi. org/10.5194/gmd-12-3241-2019
- He, B., Yu, Y., Bao, Q., Lin, P., Liu, H., Li, J., et al. (2020). CAS FGOALS-f3-L model dataset descriptions for CMIP6 DECK experiments. Atmospheric and Oceanic Science Letters, 13(6), 582–588. https://doi.org/10.1080/16742834.2020.1778419
- Lee, J., Kim, J., Sun, M. A., Kim, B. H., Moon, H., Sung, H. M., et al. (2020). Evaluation of the Korea meteorological administration advanced Community Earth-system model (K-ACE). Asia-Pacific Journal of Atmospheric Sciences, 56(3), 381–395. https://doi.org/10.1007/ s13143-019-00144-7
- Lin, Y., Huang, X., Liang, Y., Qin, Y., Xu, S., Huang, W., et al. (2020). Community integrated Earth system model (CIESM): Description and evaluation. Journal of Advances in Modeling Earth Systems, 12(8), e2019MS002036. https://doi.org/10.1029/2019MS002036
- Park, S., Shin, J., Kim, S., Oh, E., & Kim, Y. (2019). Global climate simulated by the Seoul National University Atmosphere Model version 0 with a unified convection scheme (SAM0-UNICON). Journal of Climate, 32(10), 2917–2949. https://doi.org/10.1175/JCLI-D-18-0796.1
- Rong, X.-Y., Chen, H.-M., Xin, Y.-F., Su, J.-Z., Hua, L.-J., Zhang, Z.-Q., & I, J. L. (2019). Introduction of CAMS-CSM model and its participation in CMIP6. Advances in Climate Change Research, 15(5), 540–544.
- Volodin, E. M., Dianskii, N. A., & Gusev, A. V. (2010). Simulating present-day climate with the INMCM4.0 coupled model of the atmospheric and oceanic general circulations. Izvestiya – Atmospheric and Oceanic Physics, 46(4), 414–431. https://doi.org/10.1134/s000143381004002x
- Volodin, E. M., Mortikov, E. V., Kostrykin, S. V., Galin, V. Y., Lykosov, V. N., Gritsun, A. S., et al. (2017). Simulation of modern climate with the new version of the INM RAS climate model. *Izvestiya – Atmospheric and Oceanic Physics*, 53(2), 142–155. https://doi.org/10.1134/ S0001433817020128
- Wu, T., Zhang, F., Zhang, J., Jie, W., Zhang, Y., Wu, F., et al. (2020). Beijing Climate Center Earth System Model version 1 (BCC-ESM1): Model description and evaluation of aerosol simulations. *Geoscientific Model Development*, 13(3), 977–1005. https://doi.org/10.5194/ gmd-13-977-2020

rned by the applicable Creati